Author's response to reviews

Title: Is proximity to a food retail store associated with diet and BMI in Glasgow, Scotland?

Authors:

Laura Macdonald (laura@sphsu.mrc.ac.uk)
Anne Ellaway (anne@sphsu.mrc.ac.uk)
Kylie Ball (kylie.ball@deakin.edu.au)
Sally Macintyre (sally@sphsu.mrc.ac.uk)

Version: 4 Date: 27 May 2011

Author's response to reviews:

Dear BMC Public Health,

We are pleased that our paper has been accepted in principle. The paper has been carefully proof read and typing errors corrected. We have also corrected the structure, ensured that subtitles are visible, and removed the gridlines within the tables.

The web reference in the paper is correct but it is wrong in your email so perhaps there was an error in copying and pasting as htm is missing from the end of the URL in the email (but not in the paper).

We look forward to publishing in your journal.

Best wishes, Laura Macdonald.